Author’s response to reviews

Title: A comparative cross-sectional study of some hematological parameters of hypertensive and normotensive individuals at the university of Gondar hospital, Northwest Ethiopia

Authors:
Bamlaku Enawgaw (bamlak21@gmail.com)
Nigist Adane (adanenigist@yahoo.com)
Betelihem Terefe (betch.nym@gmail.com)
Fikir Asrie (fikirie2000@gmail.com)
Mulugeta Melku (mulugeta.melku@gmail.com)

Version: 3 Date: 15 Nov 2017

Author’s response to reviews:
BHEM-D-17-00039R2
A comparative cross-sectional study of some hematological parameters of hypertensive and normotensive individuals at the university of Gondar hospital, Northwest Ethiopia
Bamlaku Enawgaw, MSc; Nigist Adane; Betelihem Terefe; Fikir Asrie; Mulugeta Melku
BMC Hematology

Dear BMC Hematology Managing Editor,

The authors thank the editor for the comment. As per the comment, we upload questionnaire and laboratory registration format as supplementary files. These additional files are cited in the method section of the manuscript. The questionnaire was validated through pre-test.

With regards
Bamlaku Enawgaw